Author's response to reviews

Title: Developing obesity prevention interventions among minority ethnic children in schools and places of worship: The DEAL (DiEt and Active Living) study

Authors:

Maria J Maynard (maria@msoc.mrc.gla.ac.uk)
Graham Baker (g.baker@sphsu.mrc.ac.uk)
Emma Rawlins (e.rawlins@sphsu.mrc.ac.uk)
Annie Anderson (a.s.anderson@dundee.ac.uk)
Seeromanie Harding (s.harding@sphsu.mrc.ac.uk)

Version: 5 Date: 14 December 2009

Author's response to reviews:

Dear Editor

The following small changes were made:

1. Carriage returns were entered so that titles are on the same page as their corresponding text

2. Dr Adamson’s title in the acknowledgements was corrected to ‘Professor’

Sincerely

Maria Maynard